Author’s response to reviews

Title: An evaluation of the DEFLIFE "self-selected" lifestyle intervention aimed at improving insulin sensitivity in people at risk of developing diabetes type 2: study protocol for a randomised controlled trial

Authors:

Grainne ODonoghue (grainne.odonoghue@dcu.ie)
Aileen Kennedy (aileen.m.kennedy@dcu.ie)
Gregers Andersen (GSA@steno.dk)
Eoin Durkan (eoin.durkan@dcu.ie)
Tanja Thybo (TTHY@steno.dk)
Margaret Sinnott (bernadette.carr@vhi.ie)
John Nolan (JJLN@steno.dk)
Donal OGorman (donal.ogorman@dcu.ie)

Version: 1 Date: 18 Sep 2015

Author’s response to reviews:

Please find attached the revised manuscript, additional files 1 and 2 and a cover letter detailing point-by-point amendments